Author's response to reviews

Title: Increased Risk of Affective Disorders in Type 2 Diabetes Is Minimised by Sulfonylurea and Metformin Combination: A Population-Based Cohort Study

Authors:

Mark L. Wahlqvist (prof-mlw@umail.hinet.net)
Meei-Shyuan Lee (mmsl@ndmctsgh.edu.tw)
Shao-Yuan Chuang (chuangsy@nhri.org.tw)
Chih-Cheng Hsu (cch@nhri.org.tw)
Hsin-Ni Tsai (970721@nhri.org.tw)
Shu-Han Yu (shuhan229@gmail.com)
Hsing-Yi Chang (hsingyi@nhri.org.tw)

Version: 7 Date: 31 October 2012

Author's response to reviews:

October 31, 2012
Dr. Lin Lee
Senior Editor
BMC Medicine

Ms. Ref. No.: 1223292856732899

Dear Dr. Lee,

Re: Increased Risk of Affective Disorders in Type 2 Diabetes Is Minimised by Sulfonylurea and Metformin Combination: A Population-Based Cohort Study

Authors: Mark L. Wahlqvist, Meei-Shyuan Lee, Shao-Yuan Chuang, Chih-Cheng Hsu, Hsin-Ni Tsai, Shu-Han Yu and Hsing-Yi Chang

The authors thank you for conveying further useful referee comment on our paper.

It was to the effect that we needed to give regard to a more appropriate diagnosis than ‘depression’ derived from the ICD-9 classification used, namely ‘affective disorders’. We have taken this advice. As you will see from the newly revised manuscript, tables and figures, we have now used the descriptor ‘affective disorders’ throughout, including the title, wherever our own work is presented. Where the literature uses ‘depression’ we have, naturally, retained the term and been circumspect about its equivalence with our study.

Insofar as we understand, there is no other outstanding issue and we commend our paper to you for publication.
With kind regards,

Mark L Wahlqvist MD FRACP
Visiting Professor
Institute of Population Health Sciences, National Health Research Institutes
35 Keyan Road, Zhunan town, Miaoli, Taiwan 35053
Tel: 886-37-246-166 ext 36366
Fax: 886-37-586-261
Email: profmlw@nhri.org.tw